Author’s response to reviews

Title: Evaluation of standard and semantically-augmented distance metrics for neurology patients

Authors:

Daniel B Hier (dhier@uic.edu;dbhier@dbhier.com)
Jonathan Kopel (jjkbqb@mst.edu)
Steven U Brint (sbrint@uic.edu)
Donald C Wunsch II (dwunsch@mst.edu)
Gayla R Olbricht (OlbrichtG@mst.edu)
Sima Azizi (sacc5@mst.edu)
Blaine Allen (bh9z8@umsystem.edu)

Version: 4 Date: 10 Aug 2020

Author’s response to reviews:

We have revised the Authors’ Contribution section to read:

Authors' contributions: Research design by DBH. Data collection by SUB, DBH, and JK. Data analysis by DBH, DCW, GO, SA, BA, and JK. Manuscript writing and editing by DBH, DCW, GO, SA, BA, SUB, and JK. All authors have read and approved this manuscript.